PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

Title (Provisional)

ASPirin for Acute Pneumonia in the elderlY (ASPAPY): Protocol of a multicentre randomised double-blind placebo-controlled trial.

Authors

Putot, Alain; Cortier, Marion; Fournel, Isabelle; Manckoundia, Patrick; Ksiazek, Eléa; Cransac, Amelie; Study Group, ASPAPY

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.